Author's response to reviews

Title: Wilkies Syndrome with an element of gut dysmotility: A Case Report

Authors:

Muhammad I Aslam (imranaslam171@hotmail.com)
Jonathan G Finch (guy.finch@ngh.nhs.uk)

Version: 2 Date: 29 May 2007

Author's response to reviews:

There are a number of matters that have been changed because the manuscript does not conform to the journal style. Following changes have been made:

Inserted an abstract into the title page.

Inserted the authors' names and affiliations between the title and abstract on the title page.

Named the first section after the abstract as 'Case presentation'.

Inserted Conclusions section at the end of the manuscript.

Inserted a Competing interests section.

"The authors declare that they have no competing interests."

Inserted an Authors' contributions section, outlining the role of each author in the study and manuscript preparation.

Detail whether written informed patient consent was obtained within an Acknowledgements section.

Inserted the words "a case report" into the title of the manuscript.

Inserted a figure title and legend section after the references.